Author’s response to reviews

Title: Association between gene polymorphisms of voltage-dependent Ca2+ channels and hypertension in the Dai people of China: a case-control study

Authors:

LiFan Huang (huanglifanhd@hotmail.com)

Yan Chu (30104096@qq.com)

Xiaojin Huang (huangxiaojin@imbcams.com.cn)

Shaohui Ma (shaohuima70@hotmail.com)

Keqin Lin (linkeqin@imbcams.com.cn)

Kai Huang (congzi@imbcams.com.cn)

Hao Sun (sunhao@imbcams.com.cn)

Zhaoqing Yang (zyang@imbcams.com.cn)

Version: 3 Date: 19 Feb 2020

Author’s response to reviews:

Dear editor Dr. Gunadi

On behalf of all the contributing authors, I would like to express our sincere appreciations of your suggestions concerning our article entitled “Association between gene polymorphisms of voltage-dependent Ca2+ channels and hypertension in the Dai people of China: a case-control study” (Manuscript No.: MGTC-D-19-00609R1). We have addressed the editor’s suggestions to best of our abilities, and revised text to meet the BMC Medical Genetics style requirements. We hope this meets your requirements for a publication.

Point-by-point responses to the nice associate editor are listed below this letter:

1. Availability of data and materials
   We have uploaded another Supplementary document which included all the raw data of our study. We hope this can meet the requirements for a publication.
If there are any other modifications we could make, we would like very much to modify them and we really appreciate your help. We hope that our manuscript could be considered for publication in your journal. Once again, thank you very much for your suggestions.

Yours sincerely,
Hao Sun

Hao Sun

The Department of Medical Genetics, Institute of Medical Biology, Chinese Academy of Medical Sciences & Peking Union Medical College
935 Jiaolin Road, Kunming, Yunnan 650118, China.
Tel: +86-871-6833 4872, Fax: +86-871-6833 4188;
Email: sunhao@km.cams.cn; sunhao@imbcams.com.cn